Author's response to reviews

Title: Purification, Characterization and Anti-aging Capacity of Mycelia Zinc Polysaccharide by Lentinus edodes SD-08

Authors:

Liqin Wang (diandianmeng@126.com)
Cuiqin Wang (wangcuiqin@163.com)
Xia Gao (gaoxia@163.com)
Nuo Xu (xunuo@163.com)
Lin Lin (linlin@163.com)
Huajie Zhao (zhaohuajie@163.com)
Shouhua Jia (shjia@sdu.edu.cn)
Le Jia (ourlaboratory@163.com)

Version: 7 Date: 19 March 2015

Author's response to reviews: see over
Dear editor,

We would like to thank you very much for giving us an opportunity to revise our manuscript again, and the positive and constructive comments which are all valuable and very helpful for revising and improving our paper. A new version of our manuscript has been modified according to your suggestions, and all the changes have been marked in red in the revised manuscript. The following is a point-to-point response to the comments.

**Editorial comments:**

1. Thank you for including an ethics statement in your manuscript. However, we would ask you to confirm that your study was submitted to, and approved by, your institutional ethics committee. Please amend your ethics statement so that this is clearly declared.

   √ The ethics statement had been amended in the revised manuscript.

2. Please include a 'Methods' sub-section in your Abstract, in accordance with our formatting guidelines.

   √ The sub-section of 'Methods' had been added in the abstract.
We have revised this manuscript according to your comments and suggestions carefully, and the rewritten paper has been resubmitted to your journal. We look forward to receiving your positive response.

Yours sincerely,

Le Jia

College of Life Science
Shandong Agricultural University
Taian, Shandong
271018, PR China.

E-mail address: ourlaboratory@163.com